Author's response to reviews

Title: Cognitive Behaviour Therapy in Medication-Treated Adults with ADHD and Persistent Symptoms: A randomized controlled trial.

Authors:

Brynjar Emilsson (brynjare@landspitali.is)
Gisli Gudjonsson (gisli.gudjonsson@kcl.ac.uk)
Jon F Sigurdsson (jonfsig@landspitali.is)
Gisli Baldursson (gislib@landspitali.is)
Emil Einarsson (emile@landspitali.is)
Halldora Olafsdottir (haldola@landspitali.is)
Susan Young (susan.young@kcl.ac.uk)

Version: 4 Date: 20 July 2011

Author's response to reviews: see over
Dr Sabina Alam,
Senior Scientific Editor,
BMC-series journals,

Dear Dr Alam,

Ref. MS:1540603760528406
Cognitive Behaviour Therapy in Medication-Treated Adults with ADHD and Persistent Symptoms: A randomized controlled trial.

Following acceptance of the above manuscript for publication in BMC Psychiatry, please find attached the latest version. We have made the following formatting changes at the proofing stage, as requested:

Major revisions (we require the author to make these changes)

Abstract: Please add trial registration information in the submission system.

RESPONSE: This information is included on the online submission system.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Affiliations: Please include country for affiliation 3.

RESPONSE: This has been done (Iceland).

Comments: Please remove post-it style comments. Eg: Insert figure 1 here.

RESPONSE: These have been removed throughout.
Highlighting/tracking: Please remove from the manuscript.

RESPONSE: Any highlighting/tracking/coloured text has been removed.

Tables: Please remove visible vertical lines from the tables.

RESPONSE: This has been done.

Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures

RESPONSE: These have now been cropped as closely as possible.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

RESPONSE: This has been completed.

We look forward to seeing this article in press.

Yours sincerely,

Dr Susan Young.